////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-OptionalInt]]++OptionalInt++::
* ++[[painless-api-reference-OptionalInt-empty-0]]static <<painless-api-reference-OptionalInt,OptionalInt>> link:{java8-javadoc}/java/util/OptionalInt.html#empty%2D%2D[empty]()++ (link:{java9-javadoc}/java/util/OptionalInt.html#empty%2D%2D[java 9])
* ++[[painless-api-reference-OptionalInt-of-1]]static <<painless-api-reference-OptionalInt,OptionalInt>> link:{java8-javadoc}/java/util/OptionalInt.html#of%2Dint%2D[of](int)++ (link:{java9-javadoc}/java/util/OptionalInt.html#of%2Dint%2D[java 9])
* ++[[painless-api-reference-OptionalInt-getAsInt-0]]int link:{java8-javadoc}/java/util/OptionalInt.html#getAsInt%2D%2D[getAsInt]()++ (link:{java9-javadoc}/java/util/OptionalInt.html#getAsInt%2D%2D[java 9])
* ++[[painless-api-reference-OptionalInt-ifPresent-1]]void link:{java8-javadoc}/java/util/OptionalInt.html#ifPresent%2Djava.util.function.IntConsumer%2D[ifPresent](<<painless-api-reference-IntConsumer,IntConsumer>>)++ (link:{java9-javadoc}/java/util/OptionalInt.html#ifPresent%2Djava.util.function.IntConsumer%2D[java 9])
* ++[[painless-api-reference-OptionalInt-isPresent-0]]boolean link:{java8-javadoc}/java/util/OptionalInt.html#isPresent%2D%2D[isPresent]()++ (link:{java9-javadoc}/java/util/OptionalInt.html#isPresent%2D%2D[java 9])
* ++[[painless-api-reference-OptionalInt-orElse-1]]int link:{java8-javadoc}/java/util/OptionalInt.html#orElse%2Dint%2D[orElse](int)++ (link:{java9-javadoc}/java/util/OptionalInt.html#orElse%2Dint%2D[java 9])
* ++[[painless-api-reference-OptionalInt-orElseGet-1]]int link:{java8-javadoc}/java/util/OptionalInt.html#orElseGet%2Djava.util.function.IntSupplier%2D[orElseGet](<<painless-api-reference-IntSupplier,IntSupplier>>)++ (link:{java9-javadoc}/java/util/OptionalInt.html#orElseGet%2Djava.util.function.IntSupplier%2D[java 9])
* ++[[painless-api-reference-OptionalInt-orElseThrow-1]]int link:{java8-javadoc}/java/util/OptionalInt.html#orElseThrow%2Djava.util.function.Supplier%2D[orElseThrow](<<painless-api-reference-Supplier,Supplier>>)++ (link:{java9-javadoc}/java/util/OptionalInt.html#orElseThrow%2Djava.util.function.Supplier%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Object,Object>>++
